



# Article Cybernetics Approach Using Agent-Based Modeling in the Process of Evacuating Educational Institutions in Case of Disasters

Ștefan Ionescu, Ionuț Nica \* and Nora Chiriță

Department of Economic Informatics and Cybernetics, Bucharest University of Economic Studies, 0105552 Bucharest, Romania; stefion09@gmail.com (Ş.I.); nora.chirita@csie.ase.ro (N.C.) \* Correspondence: ionut.nica@csie.ase.ro; Tel.: +40-728-111-808

Abstract: In the context of an emergency, evacuating people from a location in the shortest possible time is essential, as is the high degree of safety that people should expect when evacuating. Lately, in Romania there have been more and more fire events generated by different causes. This article will use agent-based modeling to simulate an emergency evacuation model in NetLogo. The model has been used to perform and analyze various scenarios. With the help of NetLogo, we managed to perform 400 simulations with the evacuation of 180 people (students, teachers, and non-teaching staff) based on which we developed several recommendations to streamline the evacuation process in order to reduce the possibility of death. The present research will help to identify the evacuation. The model that was used in this research took into account the individual particularities of the people taking part in the evacuation, emphasizing the effects that form in a crowd of people when evacuating; effects such as the funnel effect, which is caused by the formation of bottlenecks around narrow areas. All these things are part of the analysis of the measurement of entropy of the exhaust system, a problem that has captured all of the specialists' attention. Finally, solutions have been proposed to improve evacuation time in case of disasters.

**Keywords:** cybernetics approach; agent-based modeling; NetLogo; evacuation process; human behavior simulation

#### 1. Introduction

Over time, the problems generated by evacuation in cases of emergencies or disasters have attracted a large audience because such events are uncertain, and can occur anywhere and at any time, causing multiple damage, both human and material. In the last century, 13 major earthquakes were registered in Romania, which affected more than 400,000 people and resulted in over 2000 deaths [1]. Romania has been facing increasingly frequent extreme weather events in recent years-especially as Bucharest ranks third in terms of the fastest growing temperatures among the 58 largest cities in the European Union. According to information published by the World Bank, these disasters disproportionately affect over 23% of Romania's population living in poverty, both in terms of relative financial impact and longer recovery time. In the face of growing risks of disasters and climate change, Romania is focusing on developing sustainable projects and capitalizing on innovative technology to improve the way risk information is collected, analyzed, and communicated to build a strong basis for disaster risk management. This is an important point of view in terms of probability-impact and the severity of disasters, whether natural or human-caused. The World Bank and the Global Facility for Disaster Reduction and Recovery (GFDRR) collaborate with the Government of Romania to facilitate solutions for risk information, such as these, as well as other innovations aimed at reducing risks and improving emergency services. For example, the creation of a new risk assessment tool for multiple types of natural hazards at the national level (Ro-Risk) will provide all ministries with information needed to invest in risk reduction and help decision-makers



Citation: Ionescu, Ş.; Nica, I.; Chiriță, N. Cybernetics Approach Using Agent-Based Modeling in the Process of Evacuating Educational Institutions in Case of Disasters. *Sustainability* 2021, *13*, 10277. https://doi.org/ 10.3390/su131810277

Academic Editor: Ashraf Dewan

Received: 21 August 2021 Accepted: 13 September 2021 Published: 14 September 2021

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). understand the economic impact and financial disaster. The World Bank also supports the modernization of emergency infrastructure across the country, ensuring that more than 3000 first responders can rely on safer and more resilient response facilities [1].

The VISION 2020 project will improve Romania's disaster management capacity through the purchase of new equipment and staff training, aiming to better protect its population in emergency situations and to fulfill its relevant international obligations. The funding will be used to purchase emergency equipment and search and rescue operations, including fire trucks and other special vehicles, helicopters, multi-purpose vessels, and river boats. The project also includes the acquisition and equipment of a dedicated control center. VISION 2020 is in line with Directive 2007/60/EC on the assessment and management of flood risks, to reduce their impact on people and the environment. The total investment for the VISION 2020 project is EUR 681 904 797, with the EU contribution from the Cohesion Fund being EUR 578 415 116 under the High Infrastructure Operational Program for the 2014–2020 programming period. The investment is part of the priority "Promoting adaptation to climate change, prevention and risk management" [2].

In a world where the imprint of conflict and disaster is constantly expanding, helping humanitarian action is a key pillar of EU external action and an important part of its capacity to project its values globally. However, humanitarian aid is currently facing an unprecedented set of challenges, amplified by the COVID-19 pandemic. Humanitarian needs have reached a record high, largely due to the resurgence of state conflicts, combined with the impact of climate change, environmental degradation, global population growth, and poor governance. Moreover, the gap between humanitarian needs and available resources worldwide is widening. The basic rules and principles are challenged as have rarely been the case so far, making the necessary assistance more difficult and dangerous. More generally, the EU will continue to promote the development and adoption of innovative solutions that ensure more efficient, cost-effective, greener, and more climate-friendly aid [3,4].

Chen et al., in their research on assessing emergency shelter demand using POI data and evacuation simulation, state that in the field of disaster relief, determining the exact spatial distribution of the population is a challenge [5].

The risks and vulnerabilities generated by natural hazards are gaining the attention of specialists more and more. For example, Dewan developed a conceptual framework based on the risk literature on Cutter's place hazard theory with entries from recent literature. In this context, the danger is seen as a threat that has the potential to overwhelm people, property, and the environment. It is a pre-existing condition that can turn into a catastrophe depending on the influence of exogenous and endogenous factors [6]. On the other hand, Lee et al. consider that pre-planning of shelters and escape routes is the basis for efficient and safe post-flood management. They have developed a hierarchical model for selecting the location of emergency shelters in preparation for immediate, short-term, and long-term floods [7].

In the context of an emergency, evacuating people from a location in the shortest possible time is essential, as it is the most important safety measure that people should benefit from when evacuating. The probability of having or not having human lives involved in this process, that of evacuation, leads to the realization of the most accurate predictions of the evacuation plans of buildings or open spaces [8].

The new rules imposed by the COVID-19 pandemic also brought important changes in terms of building sustainability certifications. The mandatory practices and requirements for the sustainable certification of a project will be quickly adapted to the new context. For example, a criterion has been introduced in the US LEED (Leadership in Energy and Environmental Design) [9] certification scheme that provides for measures to increase safety in terms of air quality, while in BREEAM Projects [10], a credit of a potential of 100 is granted for residential project areas that have spaces for work from home. However, in recent years, in Romania there have been many disasters caused by fires in buildings, public institutions such as hospitals, which has led to both human casualties and financial losses.

Recently, Romania has felt a large influx of events within its borders, which has led to an increase in the density of people in a certain area at a certain time [11,12]. In these circumstances, a well-designed escape plan, with the best possible values for the characteristics mentioned in the previous paragraph, crucially has to reduce the number of potential victims as much as possible. We can see the devastating effects of an inefficient evacuation by taking as an example the tragedy that took place on 30 October 2015 in Club Colectiv, located in Bucharest, where, following a fire, 64 people lost their lives and another 185 people were injured, 146 of them being hospitalized. This case will remain for a long time a significant moment in Romania's history, being the worst accident that our state has had in recent years [13]. Thus, in order to avoid this type of accident, a well-developed emergency evacuation plan is imperative.

In fact, the problem of emergency evacuation is basically a problem of detecting the optimal or shortest route. The researchers used a wide range of algorithms to solve this situation, algorithms such as Dijkstra's or optimizing an ant colony [14]. Moreover, the influence of cybernetics is present even in this example. The current object of study of cybernetics is the complex adaptive system and these systems are found everywhere in the environment. It is important to understand these systems because their analysis proposes other perspectives. A complex adaptive system (CAS) is a complex macroscopic collection of relatively similar and partially connected micro-structures, formed to adapt to the changing environment and to increase its ability to survive as a macrostructure [15].

Mauboussin et al. [16], in their Harvard Business Review article, use the ant colony as an example of a CAS in nature. They aim to build a mound. Thus, it could take thousands, maybe millions of ants to build a large mound. Interestingly, ants do not start with a well-developed plan about what the mound will look like, but simply evolve and adapt along the way. Somehow, these thousands or millions of ants work together as a team that organizes itself, without too many indications. When the mound is finished, it may not be a perfect-looking structure, but it fulfills the purpose for which it was created and will evolve as needed to adapt to the general environment of which it is a part. What is important is that complex adaptive systems have been found in many natural environments since the beginning of time. What has evolved is not the systems themselves, but our understanding of them. We need to take a similar approach to project management, whether we are talking about managing a company or a disaster evacuation plan, instead of thinking of projects as a linear process in which simple rules of action apply cause and effect, we need to think of them as a much more dynamic organism [15,16].

The sustainable development of Romania must also take into account the development of the educational system. Moreover, cybernetics as a science studies the control and communication between systems, and the national economy seen as a cybernetic system or complex adaptive system is also formed by the educational system. The environment is constantly changing and people's behavior in uncertain situations is changing and is influenced by many aspects. It is difficult to predict how people will react in conditions of stress and uncertainty. This article is also based on human behavior in the conditions of a disaster such as fires or earthquakes, Romania being a country where such events occur more and more often. Disaster laws or regulations alone are not enough to protect us from these phenomena. The limited nature of human resources in educational, medical, and other institutions is well known in Romania and a management of these resources is very important.

However, it is impossible for these processes to take into account the human factor. The psychological, emotional factor or inter-human relationships are just some of the examples that can lead to a chosen route different from the one predicted by the model. Thus, these factors are essential for an evacuation time as short as possible, with the evacuation problem being a more complicated process than it seems.

Nowadays, in order for the simulation of an emergency evacuation to be as efficient as possible, the simulation's model must impersonate reality as well as possible. Obtaining the most authentic data is a challenge as they can be extremely diverse when it comes to human behavior, which leads to an increased difficulty in developing the model [17,18]. At the same time, these simulations are an efficient procedure for estimating evacuation times depending on various factors, variable or not. Thus, in the field of evacuations, researchers often use virtual models to come up with a conclusive and plausible answer.

The dynamics and predictability of space in evacuation situations is an approach to the simulation that Lu et al. [19] focused on in their article, noting that space can be altered in the event of a crisis. Another model that highlights the variability of factors and their importance for simulations is designed by Wang et al. [20,21], who used microscopic modeling to perform simulations that had as a reference the individual behavior and characteristics of each agent.

Observing and taking into account the current limitations of the aforementioned simulations, this paper aims to analyze the emergency evacuation of a middle school in different situations, taking into account the particularities of individuals who are evacuated. The study is based on the "Regina Maria" Gymnasium School, located in Sector 6 of Bucharest. In order to illustrate as accurately as possible the importance of a cybernetics approach and the proposal of the case study, we considered relevant the use of data as close as possible to reality, so we obtained the information needed for the case study from the above school. Empirical data suggest that one of the main causes of death is the irrational behavior of the mass of people when they panic, given the situation they are in [22]. Therefore, the purpose of this work is to create a model that is as close as possible to reality, appreciating the dynamics of space and without neglecting human features, which often make their presence felt during the total evacuation time of a location.

The paper is organized as follows: Section 2 provides some information about other research in the area of interest of this article, Section 3 describes the context of emergency evacuation, Section 4 presents the basic notions necessary to understand agent-based modeling, Section 5 is based on the design of the case study and the presentation of the main objective of the paper, Section 6 presents the simulation results performed in the case study, Section 7 s presents solutions that can improve the evacuation time, and in the last section were issued the conclusions and some discussions regarding the results obtained and the constraints of the case study.

#### 2. Literature Review

Analyzing the existing literature, several elements were observed that influence the individual behavior of agents. For example, Guan and Wang [23] highlight the cooperative phenomenon. They believe that knowing the evolution of cooperation has an essential role in the evacuation process. Another important factor in the evacuation process, found in the literature, is competition. Wu et al. [24] performed an experiment on the evacuation of mice from a container in conditions of high competition. Their article is based on a case study of non-human creatures, mice. They are evacuated according to different scenarios built on several variables such as the location of the exit and the distance between them or the congestion of the area.

Evacuation is characterized by the immediate and rapid movement of people from the location of a threat or disaster to a safer destination. Disasters happen every day all over the world, and these dangerous, life-threatening events force people to move immediately from a dangerous area. Disasters can be classified as natural disasters or human-caused disasters.

In 2005, Wolshon et al. [25] presented a number of hazards that require evacuation and emphasized that some evacuations, especially in the case of natural disasters, can only be carried out after disasters have occurred. They classified disasters into natural events such as earthquake, volcanic eruption, tsunami, tornado, wildfire, flood, and hurricane, and human-caused events such as terrorist attack, chemical release, dam failure, and nuclear power plant accident. In the article written by Wolshon et al. [25] they deals with ways of evacuation and transport that are found in the conditions of the occurrence of events, such as disasters. They propose several methods to achieve the goals of their research. An important aspect for our article is that they consider in their article that law enforcement agencies have a very significant role in the evacuation of educational institutions.

Other important factors that are taken into account in the evacuation process, but not limited to them are: social background [26], age [27], the degree of selfishness [28], the presence or absence of guidance or authority figures [29], the occurrence of herd behavior [30], and emotions [31]. All these factors are very relevant to emphasize that an accurate prediction of the evacuation process of an institution is quite difficult to obtain because human behavior cannot be fully predicted. On the other hand, there is still no way to classify these factors in terms of the importance or degree of appearance in people's decision-making choices. However, we could control certain factors, such as the laws and rules given by the authorities and respected by the institution.

The movement of masses of people to or from a certain location is an action that will often lead to the so-called "strait" (or bottleneck), a term that can be better described by a narrowing area (temporary or not) of the route to a certain location, resulting in an increase in the density of people around that area, leading to a situation in which the movement of individuals will be much slower, sometimes even stopped. At the same time, the appearance of these bottleneck effects has a major impact on both the safety of a mass of people during an emergency evacuation, as it creates a crowded space with many panicked people, who can act in an irrational way, and total evacuation time.

In 2005, Vermuyten et al. [32] analyzed the problem of optimizing a road facing a bottleneck, the optimization being in fact given by the largest number of people who can cross that strait per second, and in the case of evacuation problems, the optimization is given by the release time. The authors believe that in order to ensure the safety and comfort of people, it is necessary to design pedestrian facilities and good crowd management. At the same time, the analysis of pedestrian and dynamic evacuation is very complicated because the large number of people and the complex interactions between them are nonlinear, and psychological behavior and external factors are of great importance.

Thus, taking into account the above study, Duives et al. [33], through their paper, treat the movement of masses of people (according to Figure 1) and its effects as the main features of their model, using several possible displacements of the crowd, described in a tree, which have effects which lead to what is called "self-organization of the crowd", which is modeled on the following forms: string formation; stop and go waves; turbulence; herd; zipper effect; the faster, the slower. Numerous studies whose model started from "string formation" as a method of self-organization of the crowd have shown that it is among the most effective, due to congestion of the group, with the opposite pole being the "stop and go waves" due to the fact that individuals need some time to adjust to speed.



Figure 1. Cases of movement of the crowd of agents [33].

Regarding the variability of the human factor in the situation of an emergency evacuation, it is presented in a very complex way. The variability of the human factor is outlined by the possibility that some of the individuals have a disability, which would lead to a much slower evacuation of the person, but also by the psyche of each person, which will shape their behavior in such situations. For example, a study by Wood [34] showed that when people faced a fire in an enclosed space, 15% of them first wanted to put out the fire, 13% started the fire alarm, and only 9.5% started evacuating the next moment. Another researcher, Sime [35], mentioned that people tend not to be aware of the danger immediately, wanting to validate it for the first time, instead of evacuating immediately. However, regardless of the cause of the emergency evacuation, people tend to become irrational, illogical in their actions, and disorganized, which leads to an exponential increase in panic. A representative perspective, which expresses the factors that can induce panic in individuals in the event of an emergency evacuation (according to Table 1), is suggested by Proulx [36]. Proulx classifies these factors according to occupant characteristics, building characteristics, and fire characteristics. Dr. Guylene Proulx, in her article, argues that although most buildings are equipped with adequate fire safety systems, problems often arise because these systems are not made to react to the way human agents behave in those buildings. People tend to ignore fire alarms in shopping malls, museums, or airports, or they have installed stops to keep the doors open during the day, so all these factors have a negative impact on an effective evacuation without loss of life. The author of the article argues that human behavior varies depending on three major characteristics: the occupant characteristics; the building characteristics, and the fire characteristics. Dr. Proulx argues that one solution to these is to consider the interaction of all these factors in order to better understand human behavior [36]. Here comes, again, the cybernetics analysis that takes into account both the internal environment analyzed, but also external factors.

Occupant Characteristics		<b>Building Characteristics</b>		<b>Fire Characteristics</b>	
	Gender Age		Residential Office		Flame Smoke (color, thickness) Deflection of wall, ceiling, floor
Profile:	Ability	Occupancy:	Hospital	Visual cues:	
	Limitation		College and University		
Knowledge and Experience:	Familiarity with the building Past fire experience Fire safety training Another emergency training	Architecture:	Number of floors Location of exists Complexity of space Visual access	Olfactory cues:	Smell Burning Acrid smell
Condition at the Time of Event:	Alone vs. with others Active vs. passive Alert Under Drug-Alcohol-Medication	Activities in the Building:	Studies Working Watching a show Sleeping	Audible cues:	Cracking Broken glass Object falling
Personality:	Influenced by others Leadership Negative toward authority Anxious	Fire Safety Features:	Fire alarm signal Voice communication system Trained staff Fire safety plan	Other cues:	Heat
Role:	Visitor Employee Owner Student				

Table 1. Factors influencing human behavior in the event of an emergency evacuation [36].

In addition, the decision taken by individuals in such a context, that of evacuation, differs from day to day. Usually, under pressure, people tend to think that the decision needs to be made as soon as possible, even if the information on which the choice should be made is limited or too numerous. Therefore, behavior such as moving away and running away from the incident may seem, from the outside, a disorganized decision, but for those who evacuate, it is possibly the most rational response if their integrity is endangered.

In fact, many studies treat the idea of panic as a defining factor in the designed models, trying to simulate as well as possible the situations described by them, but due to the behavioral and mental peculiarities of each individual, creating a model that accurately describes real life based on the existence of panic is almost impossible.

As mentioned, this paper aims to address the evacuation of students from a school in an emergency situation, with a scientific basis on concepts such as the movement of masses of people in crowded situations or human behavior when faced with a situation of crisis. Thus, by carrying out the proposed research, the result of the analysis will be followed, which will take place as a result of the interweaving of the previously mentioned terms in order to elaborate several pertinent conclusions.

#### 3. Presentation of the Evacuation Context

#### 3.1. Location of Exits

An efficient evacuation plan is tailored so that there is a strategic location of the exits from the respective facility, so that each individual obtains an optimal evacuation time. However, even if an evacuation plan often revolves around human behavior, as it should, it does not address situations where obstacles arise within the evacuation site [37]. In this way, an imbalance is created in exit placement, as not all of them will be able to be used, which implies a less optimal evacuation time or even the injury of the people who are released.

A study attesting to the influence of the location of the exits inside an enclosure was carried out by Heba et al. [37], whose simulations could be better visualized using a system based on cellular automata and multi-agent technology. Four components made up this system: the plan generator, the choice of technique (depth-first-search/DFS), monitoring operations, and visualizing performance, and simulations were performed based on four equal rooms, each with four outputs that were located differently.

After performing the simulations, taking into account the human behavior, it was concluded that the evacuation of people could not be evenly distributed in any of the four possible circumstances, even if there was a greater balance between exits and different areas of the room in the case of exits placed diametrically opposite and the outlets placed on a single wall [37]. In order to establish a better dispersion of people between the exits, for the situation in which the second room was used, that of the adjacent exits, it has been shown that a new exit near Exit 1, or one positioned in a diametrically opposite way to Exit 1 has a lower density of people around the exits, the percentages being 27% and 33%, respectively, of the total population in the room [37]. However, many limitations are identified when it comes to studying the effect that position of the exits from a particular location has, due to the fact that few articles in the literature refer to this topic.

#### 3.2. Emergency Exits

It is well known that emergency exits are important because they provide a safe and clear way to evacuate a location in situations that endanger the lives of those inside it. Local authorities responsible for mitigating the situation, such as the police or fire brigade, can use those emergency exits to enter the building and help any people who have not yet managed to reach a safe place. Thus, an emergency exit is defined as a continuous, unobstructed route leading to a safe area, as noted by the US Occupational Health and Safety Administration [38]. This department defines emergency exit as a concept that encompasses the following three characteristics:

- Access route—leads to the emergency exit and must be at least 70 cm wide at any point;
- Exit—is usually separated from other areas, to provide a safe way to reach the exit of the building;
- Exit the building—leads directly to a safe space.

It should be noted that these access routes and emergency exits must not be blocked, in order to reduce the evacuation time as much as possible, but due to unpredictable events, obstacles may appear around them, which may even block the emergency exits, a scenario in which panic will break out among the masses.

#### 3.3. Chain of Command in Case of Evacuation

A chain of command is an essential aspect of the organizational structure that aims to delimit and distribute a responsibility to each person, depending on the degree occupied in a certain location (e.g., visitor, director, etc.). During an emergency evacuation, it is mandatory that order has to be maintained. If it is not kept as efficient as possible, people evacuating or coordinating them may be harmed or, often times, may lose themselves organizing people [39]. The person who usually handles the emergency in the event of an evacuation is usually a qualified person in this matter, who is obliged to inform the other individuals about the steps to be followed. In the case of a school, the responsibility for managing a crisis situation often falls on the shoulders of the person with the highest authority at that time, i.e., the school principal, or the teachers (in special circumstances, if the school's auditory information system does not work). This approved entity must follow well-defined protocols and provide precise guidance, being responsible for how the discharge will be carried out. Given the fact that in the case of evacuation of a school, children can panic very quickly, thus becoming disorganized, and the following lines include some of the responsibilities of such an authoritarian person:

- Checking the rooms (preferably the nearest one, if there is a major incident) to identify people who are stuck or unable to evacuate;
- Know who may need support during the evacuation and how these people can be helped;
- Coordination of emergency activity;
- To know the architecture of the school, and the evacuation routes most suitable for the context;
- Check that students are distributed to safe areas after evacuation.

#### 3.4. The Role of Evacuation Simulations

Existence of evacuation simulations in case of fire/earthquake, etc., proved to be of major importance in the reaction time of the people involved. Thus, through these simulations, certain situations can be predicted that, in real cases, even when life is endangered, people learn how to respond to unforeseen events.

As early as the middle of the last century, most school systems began to implement, frequently, fire simulations, in which all those present in the school practiced evacuation from school, reaching a safe area [40]. In areas vulnerable and susceptible to natural disasters, many schools perform evacuation simulations for contexts such as earthquakes, tsunamis, tornadoes, wildfires, and terrorism [41].

It turned out that due to fire simulations, there was a decrease in injuries or deaths that occurred during such an incident, which was common in the 19th century and the first half of the 20th century [42].

However, even if schools are prepared to face such unpleasant situations, there is a factor that intervenes when everything is real, and that factor is emotion. During a seminar given by Lt. Col. Dave Grossman [43], he talked about the fact that a human's most important survival mechanism is their brain, but only when they are prepared as such. Undoubtedly, the incredible pressure that a crisis situation can bring can have extreme effects on the human body. Unfortunately, when they are put in a circumstance that

could lead to their injury, people become confused, panic, and fail to complete activities that present a small degree of difficulty. Thus, in order to improve the simulations, those responsible for the response to an emergency evacuation from each school must explore ways to include cognitive thinking [43]. These improvements should encourage teachers, students, and school staff to think quickly when emotions are high, and some strategies that propose these improvements are:

Encouraging role change:

Simulations can be organized to include role change. Thus, for example, the participants in the simulation will be given the opportunity to decide whether to evacuate, to be the ones to trigger the alarm, to barricade the classroom doors, to charge an armed individual, and so on.

Incorporation of real scenarios:

Simulations can be organized in such a way that surprise elements are added, such as blocking some exits in the last moment of the evacuation to test the reaction skills of the participants, etc.

> Interweaving the simulations with the theoretical exercises discussed:

Unusual simulations can be organized in which to discuss various theoretical but real situations, which participants can face. Compared to the method used in the previous sub-item, it includes several situations, being more discussed, such as not necessarily having a well-established escape route in case of their occurrence.

#### 4. Conceptual Dimensions of Agent-Based Modeling

In the literature, there are many definitions formed by various researchers regarding agent-based modeling, which are more or less similar, but their meaning is to some extent the same. Thus, the significance of an agent-based model is given by a computerprogrammed model, using various mathematical or statistical formulas, which aims to capture the evolution and behavior of one or more individuals, but within a certain environment, with conditions well established, depending on what we want to simulate or study [44].

The advantage that agent-based modeling offers is represented by the visualization of the results and the process by which they were reached using an interactive form, not an abstract one, even if the starting point in solving the problem is an abstract, theoretical structure, such as a formula or an indicator. For example, one can also observe how a certain agent changes during a simulation, which is often associated with real-life entities; video games such as "The Sims" or "SimCity" state what has been said before, because a person can transpose in the virtual environment, interacting with other people or with the game environment [45].

Currently, there are many main types of agents that can be an agent-based application. Their classification, nowadays, is quite detailed and segmented, using various criteria according to which the hierarchy is made, such as: the functions that are performed by agents, their properties, their type, their behavior, etc.

As Bradshaw et al. [46] mentioned in their article, several types of agents are highlighted when they are related to their properties, namely:

Cooperative agents: agents who synchronize their actions and, in order to achieve common goals, they negotiate;

Autonomous agents: tend towards a certain goal, acting in such a way as to achieve it, without the need for intervention or confirmation from the user. They are also called proactive agents;

Adaptive agents: flexible agents that adjust in a dynamic way, learning from and about their environment. They can adapt to the unforeseen, but also to changes;

Reactive agents: agents that are triggered by the occurrence of events, being sensitive to the context of the environment. They are able to act and feel;

Mobile agents: agents who follow a certain route, moving. This movement can be performed in both the real and the physical environment;

Interactive agents: their responsibility is to interact with legal systems, people, information sources, or even other agents;

Social agents: agents who are cooperating with agencies or people who have common goals;

Agents with personality: they have particularities that are found among people, such as emotions, values, cultural affiliation, etc.;

Intelligent agents: agents that mimic the defining characteristics of human intelligence, such as introspection, meditation, etc.

When agents relate to the functions they perform, the classification is as follows [46]: Information agents: by collecting information from several sources, they forward the information to many other sources;

User interface agents: agents that are in a correspondence relationship with people, the communication being undertaken through several types of interfaces;

Reactive agents (or actors): agents who perform certain operations in an independent way, but are nevertheless influenced by the events and messages in the environment that were triggered before the operation;

Mediating agents: agents responsible for the distribution of any resource between people and/or various types of agents.

When discussing the behavior of agents, it is taken into account that they must best represent human behavior. Therefore, for the integration of human and social behavior in the results offered following the simulations that are performed, three types of behaviors of agents are outlined, ranked according to their complexity [47].

(a) Behavior during movement:

This type of behavior corresponds to the simplest action that an agent can take, namely moving. Thus, an agent can walk or run forward, stop, turn, or follow a previously traveled route, and travel at a certain point in time depends on a well-defined decision or rule [44].

(b) Behavior during change of direction:

Used very often in artificial and robotic intelligence, this type of behavior is essential in the path followed by an autonomous agent in its virtual environment, because the journey must be undertaken in a more realistic way [47]. Thus, more complex simulations can be performed, which present more interesting results than those that are based only on behavior during movement. Next, some particularities that underlie this type of behavior will be described, and starting from them, much more complex situations can be created [47]:

- Random movement;
- Exploration—has the role of guiding an agent to a desired point. When such a point is detected, the agent adapts their speed and orientation to that location;
- Negotiation—due to this feature, agents can exchange information with each other, and can even reach certain agreements;
- Collision avoidance—gives agents the ability to choose unobstructed routes/other agents;
- > Target tracking—gives the agent the ability to track a moving agent.
- (c) Behavior in social situations:

Described by its complexity, behavior in social situations is based on the premise that there are multiple interactions between autonomous agents, for which, if they were isolated, the simulations would have felt small deviations between them, but being in a group of such agents, their results helped specialists to observe patterns of social behavior [48]. These include the herd effect or competitive social behavior, in which case individuals compete for the chance to leave a particular location.

Agents of agent-based modeling are able to provide unbalanced solutions, but also to describe complex phenomena. Often, one of the characteristics of the results achieved by an agent-based model is the emergence of self-organization. Self-organization is based on the existence of a certain model, a pattern to be followed, or on its creation from the bottom up, and the emergence can be applied to structured phenomena that include higher order functionalities [48].

Agent-based models in trading and negotiation often produce different results, with Kirman and Vriend's studies [49,50] being conducted in this matter. In their research, the financial market was abstracted, offering buyers and sellers the opportunity to form relationships with each other, a situation in which individuals had different customer behaviors. For example, customers who offered more for goods are generally less loyal to sellers than customers who offer less money. Thus, the agent's reputation and networking ("social network") are present in a very complex way, being determined and directly influenced by the interactions between them.

Agent-based models are also used to study the distribution of event dimensions [48]. When the aggregate variables are equaled by the sum of the individual variables, it means that, in more relaxed conditions, these parameters will be normally distributed. However, when there is positive feedback or when the values of particular attributes are independent, it means that the aggregate variables are able to support more extreme distributions.

A graphical representation of the event size distribution is the most used method in analyzing the aggregate results [48]. For example, according to the study conducted by LeBaron [51], when an agent-based model is built and formed on the stock market, it is possible to graph the daily fluctuations of stock prices on the stock exchange, where it is possible to observe indices of bubbles and the fall in stock prices for certain companies. In other situations, such as patterns of traffic flows, the entities that make up the model can self-organize in critical states, where even seemingly insignificant events can have a large impact on the outcome.

Agent-based modeling is represented by computer systems in which, based on wellestablished rules, agents interact with each other, taking place in an environment [52]. On the other hand, the set of rules and conditionings defined by scientists can outline predictable behavior at the micro level, with interactions between agents and their environment leading to unexpected results from experiments [53].

#### 5. Case Study: Emergency Evacuation of a School

The problem proposed in this paper is the evacuation of the ground floor of the school "Regina Maria", located in the Bucharest's 6th district, which targets the studying of the impact of various parameters in the context of an evacuation, such as obstacles or possible disabilities that children may have. With a first-floor capacity of exactly 202 people, taking into account not only the children, but also the teachers (18 people), the principal or the secretaries (4 people), and with three access ways (all for entering/leaving the school, unfortunately without any emergency exits), will be one of the parameters that the model used to perform simulations, and that intends to highlight, as well as possible, the effects of specific values for the parameters considered. The 22 people (principal, secretaries, and teachers) are considered fixed agents in the model, so they will always be present. The number that varies is that of students, i.e., 180 people.

## 5.1. School Architecture

Being built in 1970, "Regina Maria" school is a "U" shaped one, on three floors, with classrooms on both sides, the two "wings" being separated by a hall. Thus, in order to describe the architecture of the model, we chose to present the dimensions of the rooms only for the first floor, as only these are relevant for the study. Therefore, the dimensions of the rooms located on the ground floor of the educational institution are specified below. Presentation of room dimensions:

- > The size of a classroom is  $(6.5 \times 9.5)$  m<sup>2</sup>, with a maximum capacity of 30 children plus a tutor (marked with the letter A in Figure 2);
- > The size of the main hall is  $(3 \times 35.5)$  m<sup>2</sup> (denoted by the letter B in Figure 2);

- > The size of the side hallways is  $(3 \times 29)$  m<sup>2</sup> (denoted by the letter C in Figure 2);
- > The size of the teachers' office is  $(4 \times 6)$  m<sup>2</sup>, the principal's office is  $(3.5 \times 5)$  m<sup>2</sup>, and the office of the secretaries is  $(3 \times 5)$  m<sup>2</sup> (marked with the letter D, E, and F, respectively, in Figure 2).



Figure 2. Architectural sketch of the school (made by the authors in draw.io).

Output size of the rooms:

- > The main exit has the area of  $(3 \times 5)$  m<sup>2</sup> (denoted by the letter G in Figure 2);
- > The teachers' exit has the area of  $(7 \times 5.5)$  m<sup>2</sup> (marked with the letter H in Figure 2);
- > The rear exit has the area of  $(2 \times 3)$ m<sup>2</sup> (denoted by the letter I in Figure 2).

In Figure 2, we can better see the architecture of the school, which also provides a visual representation of the rooms inside it. The entrances of the school are marked with a shade of green, to differentiate themselves from the rest of the doors, which can be "intermediate" doors, which only lead to one of the three exits.

#### 5.2. Building the Model in NetLogo

Created in NetLogo, the model aims to best imitate reality in the case of an emergency evacuation. Thus, there are several factors that directly impact the evacuation time from the building and the results of the simulations; factors such as the number of people present or the physical characteristics that each of the children may have. According to the National Institute of Statistics [54], in Romania there are approximately 70,000 children with disabilities of any type (visual, olfactory, locomotory, etc.), which constitutes a percentage of 0.1% of the total number of children enrolled in a pre-university education cycle.

The process of an emergency evacuation involves a fast reaction time, with the main method of evacuation being walking. Being subjected to an evacuation, the behavioral peculiarities of individuals, such as panic, directly impacts the release process inside each individual who evacuates, triggering the mode of survival. Therefore, three behavioral characteristics were considered for the accuracy of the simulations:

- (1) Choice of exit: The evacuating agents will be aware of the three access routes of the school, and they will choose a single exit for evacuation, based on the distance criterion. Thus, an agent will choose to evacuate using the nearest of the three available exits, around which the "funnel effect" is formed.
- (2) Relative position in space: Characterized by the location of the agent in the model, this factor greatly influences the evacuation times, because a shorter distance between the exit and the agent will lead to a relatively short evacuation time. At the same time, obstacles will appear inside the school, which can be skipped or bypassed, which will influence the time of evacuation.
- (3) Travel speed: According to a study by Thiele et al. [55], an individual's ability to evacuate takes into account factors such as tone or physical features. Thus, in the model, these factors will be reflected in the speeds of the agents.

In this research, a factor that influences the speed of children who are evacuated is represented by the evacuation instructions held in schools, given the number of children who participated in these courses, as they have a higher travel speed because they know certain procedures that will help them evacuate faster, and they will be less panicked, treating the situation with greater calm. There are few studies that specify how much the speed of travel changes as a result of evacuation instructions or fire simulations, but according to a few studies conducted by Peacock et al. [56,57], it was shown that the movement speed of those who participated in such trainings increased by up to 10%, which was noticed due to the shorter evacuation times. At the same time, another factor that influences the speed of children in this work is the help and instructions provided by the tutor to the children for whom they are responsible. Based on an article by Kawasaki et al. [58], we were able to estimate an increase in children's movement speed of up to 8% around teachers, as teachers have been shown to take the initiative to evacuate due to the confusion created among children, guiding and helping them, especially those with disabilities.

Therefore, following a study by Chu et al. [59] on the emergency evacuation of people from residential areas, we assessed the average travel speeds as those in the Table 2, also taking into account the influence of the factors presented above. Furthermore, in the Table 2 is included the maximum number of people in each category that will be present in the simulation, and in the case of students with disabilities, we estimated an upper limit of 10%, taking into account the fact that the number of children enrolled in the secondary education cycle fluctuates every year, with the students with disabilities constituting a percentage of 0.1% of the total children.

**Table 2.** Average travel speeds of people participating in the simulation.

Category	Average Travel Speed (m/s)	The Presence of Teachers Around	Maximum Number of People (Absolute or Relative)
Student	1.02	(1+0→8%) * 1.02	180
Student who participated in evacuation training	(1+0→10%) * 1.02	(1+0→10% + 0→8%) * 1.02	$(0 \rightarrow 100\%) *$ Total Students
Professor	1.2	N/A	22
Student with disabilities	0.42	(1+0→8%) * 0.42	$(0 \rightarrow 10\%) * \text{Total students}$

These types of agents can be distinguished in the model made using the color they have in the simulation:

- Student—marked in yellow;
- Student who participated in evacuation training—marked in green;
- Professor—marked in white;
- Student with disabilities—marked in red.

An architectural sketch of the school was also made in the NetLogo interface, which, this time, also includes the agents involved in the analysis. Thus, Figure 3 shows the model to be used, thus providing a better overview.



Figure 3. Architectural sketch and location of agents within the model (designed by the authors in NetLogo).

The above-mentioned parameters, which can impact the evacuation times of the simulations, will be presented, together with the other parameters that can have an effect on the evacuation times, using the NetLogo interface in the Figure 4.



**Figure 4.** Parameters used in the analysis of emergency evacuation of the school (designed by the authors in NetLogo).

Below, we will describe the parameters in the Figure 4, providing more details for the parameters that have not yet been presented in the previous rows:

- Number-of-children: the total number of students of the simulation, which will be randomly distributed between classes;
- %-Children-evacuation-courses: represents a certain percentage of the total number of students present;
- %-Children-with-disabilities: represents a certain percentage of the total number of students present;
- %-Probability-of-obstacles: randomly placed, represents the density of obstacles in the total area of the first floor of the school;
- Main-exit: one of the three access roads;
- Backdoor-exit;

#### ➤ Office-exit.

The three access roads to the educational institution can be open, semi-open, or closed.

#### 5.3. Obstacles as a Major Effect during Evacuation

Obstacles are a major impediment for people who want to evacuate in emergencies, as they not only reduce the space in which people can act, but also force individuals to choose other alternative routes, which are safer in terms of their integrity but lead to a longer evacuation time. The precious seconds that are lost when the route deviates from the normal one is unfortunately one of the most important factors that decide whether the evacuee will manage to get injured or not in a safe place, and safe from danger after release.

However, according to a study by Wei et al. [60], in which the simulations were performed on the basis of a room of  $(1.6 \times 3.2)$  m<sup>2</sup>, it was shown that the evacuation times were similar, regardless of whether the obstacles were large or small, but only if they were at least 1.5 m in size.

It has also been shown that, in very few cases, the appearance of obstacles in strategic places leads to a slightly shorter evacuation time than if there were no obstacles, as the bottleneck effect is more difficult, but once formed, it is much more expensive from the escape-time point of view, compared to the obstacles themselves. However, in most situations, obstacles do not have a positive impact on those who evacuate, primarily because of the uncertain route that individuals must follow, which almost always deviates from the one that would have been followed under normal conditions, and secondly because of their possibility to block some escaping routes, in which case we can even talk about loss of human lives.

Based on studies conducted by Wei et al. [60], we were able to estimate the density of obstacles in the model proposed in this research, with the obstacles being placed between 0–25% over the surface of the first floor of the school, but also the minimum distance between an access path and an obstacle, which must be at least two meters. In Figure 5, we can see more clearly the density of obstacles in the proposed model, with a value of 25% of the area, and being a shade of purple.

#### 5.4. Proposed Scenarios Based on the Model

Creating scenarios that are reflected in the simulation results is an essential step in the veracity of the model, because there cannot be more emergency evacuation simulations performed within the same school that will lead to identical results, but rather similar ones. In reality, it is almost impossible for evacuation times to be identical under similar conditions, as the uncertainty factor must be taken into account, such as a child stumbling, or a different distribution in classrooms at the beginning of the evacuation simulation, which leads to already another scenario.

#### 5.4.1. The Baseline Scenario

The model proposed in this paper will start from a Baseline Scenario, which is exemplified in Figure 6 and for which the following values are known for the parameters used:

- During the simulation, the maximum attendance at classes is considered;
- All three access roads will be fully open and can be used by evacuees;
- A value of 30% of all students for children who have given evacuation instructions, due to the fact that they are often ignored, or the presence of children at school was low when they were held;
- Children with disabilities will be in proportion of 1% of the total students;
- There will be no obstacles.



Figure 5. Random obstacle distribution (designed by the authors in NetLogo).



Figure 6. Placing agents in the baseline scenario (designed by the authors in NetLogo).

For each scenario, 400 simulations will be performed in such a way that the results are as accurate and relevant as possible. According to Safelenics [61], it has been shown that, in the case of a school, an optimal evacuation time would be two and a half minutes; this being measured from the place where individuals start to evacuate, until they reach a "green" point, safe from danger.

# 5.4.2. Scenario I-Variability of Access Paths

The simulations based on Scenario I will rely on the openness of the access ways, because often times they were not completely open, and sometimes they were even closed, as happens in reality. To study this scenario, the parameters that will change are the main entrance and the teachers' entrance openness, with the rest of the properties remaining constant, similar to those in the Baseline Scenario. Therefore, the situations that may arise as a result of this scenario are the following:

## A. The teachers' exit will be semi-open:

The deviation from the Baseline Scenario can be seen in the Figure 7, where the access road that is delimited by the two administrative offices is partially open, by only two thirds of the width.



Figure 7. The teachers' exit: semi-open (designed by the authors in NetLogo).

B. The rear exit will be semi-open:

In the model based on the "Regina Maria" school, the rear exit will be considered as semi-open, and thus the bottleneck effect takes a more accentuated shape. Figure 8 highlights the modification of the rear access path and the deviation from the Baseline Scenario for the case from this subpoint.



Figure 8. Rear entrance: semi-open (designed by the authors in NetLogo).

#### C. The rear exit will be completely closed:

As in the previous case, in the Figure 9 will show the change and deviation from the Baseline Scenario, with the rest of the variables remaining constant.



Figure 9. Rear entrance: closed (designed by the authors in NetLogo).

#### 5.4.3. Scenario II—The Appearance of Obstacles as a Factor of Uncertainty

Performing simulations based on Scenario II will be based on the variability of obstacles, as they are an unforeseen element, which is theoretically taken into account in emergency evacuation training or courses, but it is quite difficult to implement an evacuation with obstacles, mainly because of their uncertainty of occurrence. Therefore, for the study of this scenario, the parameter that will change is the total number of the obstacles, with the rest of the properties remaining constant, similar to those in the Baseline Scenario.

Therefore, the situations that may arise as a result of this scenario are the following:

A. The appearance of obstacles on 10% of the surface of the first floor, randomly:

A percentage of 10% of the first-floor area to be considered as an obstacle during an emergency evacuation can be perceived as a unique value, but taking into account the interior structure of a school, with elements such as paintings, locker rooms, and closets, this is possible. Thus, Figure 10 highlights the surface with obstacles, while outlining the difference between this case and the Baseline Scenario.

B. The appearance of obstacles on 20% of the surface of the first floor, randomly:

As in the previous case, in the Figure 11 will show the change and deviation from the Baseline Scenario, with the rest of the variables remaining constant.

#### 5.4.4. Scenario III—Variation of the Number of Present Children

The severity of a situation that causes an emergency evacuation has as an influencing factor the total number of people inside the building at the time of the evacuation, because a small number of people will theoretically lead to a shorter evacuation time.



Figure 10. Obstacles in case of emergency evacuation (designed by the authors in NetLogo).



Figure 11. Obstacles in case of emergency evacuation - change and deviation (designed by the authors in NetLogo).

At the same time, in a school, it is often the case that students are unequally distributed between classes, because they may be absent or absent from classes, thus creating an inequality between the places occupied in each class. For example, in the context of the COVID-19 pandemic, some students may be absent for a certain period of time, if they are positively confirmed with the existing virus. Thus, considering the random seats in which students are placed within the classroom in the proposed model, to study this scenario, the parameter that will change is the total number of children, with the rest of the model properties remaining constant, such as those in the Baseline Scenario. Therefore, the situations that may arise as a result of this scenario are the following:

A. Decreasing the presence by 15% of the initial value:

A lower attendance by 15% leads to a number of 27 individuals who are absent from the course, with the total number of children present being 153 in this case. This low attendance of 85% at classes, can be caused by various reasons, such as social, family, bad weather, student illness, etc. Figure 12 provides a larger overview of the differences between this case and the Baseline Scenario.



Figure 12. Example of placing children at a 15% lower attendance (designed by the authors in NetLogo).

#### B. Decreasing the presence by 35% of the initial value:

As in the previous case, the distribution of children will be observed when their presence is worth 65% of the total capacity of seats in banks, with only 117 of them being occupied. Thus, Figure 13 represents how the minimum number of students are divided in order to simulate, with the rest of the parameters remaining constant with values identical to those in the Baseline Scenario.

The values of the parameters for which the simulations were considered for this case can be better visualized in Figure 14.

In the Figure 15 shows the bottleneck effect that was formed in the model studied during the emergency evacuation, with this factor having the same effect in all the studied scenarios, with it being in fact the reason of the increase in time when it comes to the evacuation time, given the normal sizes of the doors in ordinary situations, but relatively low when considering such an evacuation situation. The phenomenon that occurs as a result of the bottleneck effect is described as the funnel effect, where the distribution of people evacuating when they encounter a narrower place takes the form of a funnel, as can be seen in Figure 15. In fact, at the beginning of each simulation, students with disabilities will always be helped and guided by the closest person.



Figure 13. Attendance of 65% of children in classes (designed by the authors in NetLogo).



**Figure 14.** Parameter values in cases where the presence is minimal (designed by the authors in NetLogo).



**Figure 15.** Formation of the bottleneck effect in the analyzed model (designed by the authors in NetLogo).

# 6. Results

In order to perform a more accurate and relevant analysis of the model, 400 simulations were performed for each case of the scenarios previously studied. The results include the most important aspects that appear in an emergency evacuation, such as the total evacuation time and the number of injured people, with these being the main criteria based on which the severity of such an unpleasant event was established. An average of the aspects mentioned above were performed, and they are represented in Table 3 for each situation.

Situation	Total Evacuation Time (seconds)	Total Number of Injured People
-	153.363	0.023
А	153.543	0.028
В	157.823	0.038
С	184.248	0.054
А	169.323	0.063
В	188.815	0.110
А	146.304	0.021
В	137.012	0.010
	- A B C A B A	Situation         Time (seconds)           -         153.363           A         153.543           B         157.823           C         184.248           A         169.323           B         188.815           A         146.304

**Table 3.** Simulation results for each situation.

Based on Table 3, we observe that the evacuation times fall within the interval (137.012; 188.815) seconds. Considering an optimal evacuation time of a school as two and a half minutes, a fact claimed by Safelenics [61], these values are approximately in the area of that time. However, there are quite large deviations in certain situations of the proposed scenarios, with these being given by Scenario II—A, B, and Scenario I—C. Therefore,

a detailed analysis was performed on the scenarios studied to identify the reasons that led to those times.

Scenario I—Variability of access paths: A. The teachers' exit will be semi-open:

The evacuation time of this scenario was very close to that of the Baseline Scenario, so an open exit of teachers is not reflected in an evacuation time that deviates to a large extent from the optimal situation—two and a half minutes. However, the number of injured people is 28 per 1000 people, with this being most likely caused by the panic that was created in the crowd, with people being disoriented or rushing to evacuate, without paying attention to those around them.

Scenario I—Variability of access paths: B. The rear exit will be semi-open:

When the back door was ajar, there was a small increase in evacuation time, compared to the Baseline Scenario, which has a value of 157,823 s, compared to 153,363 s. Even if the difference between the two times is apparently small—four seconds—which are essential in an evacuation, and it may be that they make the difference between life and death in certain situations. We can notice that the number of injured people is 38 per 1000 people, which is a slight increase compared to the previous situation because the rear entrance has a much narrower width than the teachers' entrance, thus forming a much denser crowd around the exit than the one formed in Scenario I—A. As the density of panicked people has increased, an increase, no matter how small, of the injured people is inevitable.

Scenario I-Variability of access paths: C. The rear exit will be completely closed:

There is a large discrepancy between the evacuation times between the current situation and the Baseline Scenario, as there was a longer distance to be covered by the students to the exit, as about half of them were not able to use the nearest exit (the back one), but the main one, which involves a longer route than the initial one. At the same time, the number of injured people is 54 per 1000 people, being also influenced and determined by the panic that broke out among the people who were evacuating.

Scenario II—the appearance of obstacles as a factor of uncertainty: A. The appearance of obstacles on 10% of the surface of the first floor, randomly:

In an emergency evacuation, it is very likely that obstacles will intervene in the route chosen by those evacuating, which will leave their mark on evacuation times, but also on the individuals' integrity. Therefore, in the case of this scenario, the evacuation time increased by about 15 s compared to the Baseline Scenario, due to obstacles, which change the dynamics of students, with some choosing to jump over obstacles, and others choosing to avoid them, but regardless of the choice made, the evacuation time increases, precisely because of the deviation from the normal route, with the shortest not being able to be followed in most similar situations. At the same time, the number of injured people reaches the value of 63 per 1000 people, because in addition to the panic, the obstacles make it difficult for people to evacuate, with some of them even injuring themselves in this. An example of this can be given by the burns caused to individuals after touching hot objects, in their attempt to evacuate as quickly as possible.

Scenario II—the appearance of obstacles as a factor of uncertainty: B. The appearance of obstacles on 20% of the surface of the first floor, randomly:

An obstacle density of 20% on the surface of the first floor leads to an evacuation time of 188,815 s, being the largest of all times for the scenarios studied. It exceeds the optimal average evacuation time by 38 s, a very high value in this context, which can have disastrous effects on those evacuated, with the unhappiest scenario being the loss of human lives. Therefore, it is observed that there are 110 per 1000 injured people because, compared to the previous scenario, the space of action of those who evacuate has decreased considerably, and they have to think about the next move in a very short time, creating an imbalance between panic and integrity or possibility of self-harming them, an imbalance that dictates how each individual will evacuate.

Scenario III—variation of the number of children present A. Decreasing the presence by 15% of the initial value:

A 15% decrease in the number of people evacuating compared to the Baseline Scenario will lead to an evacuation time that is shorter by 7 s, which is also less than the optimal evacuation time of a school. However, it is observed that, although there are fewer people involved in the evacuation process and their density being lower, panic is still established among those who evacuate, bringing the number of injuries to the value of 21 per 1000 people. No matter how small the number of people evacuating, undoubtedly there will always be panic within when they know they are endangered and that they can be seriously injured.

Scenario III—variation of the number of children present B. Decreasing the presence by 35% of the initial value:

A 35% decrease in the number of people evacuating compared to the Baseline Scenario results in a total evacuation time of 137 s, much less than the optimal evacuation time of a school. However, even if the number of people involved in the evacuation process is one third lower than in the Baseline Scenario, there are still injured people; 10 per 1000 people.

According to the differences between the studied scenarios, it is easy to see what influence various factors can have on the evacuation time, but also the complexity that emergency evacuation shows in creating a model that includes all the elements that impact evacuation times. Obstacles, how wide the entrances are, the number of people present, or panic are just some of the properties that make their mark on both the evacuation time and the number of injured people.

#### 7. Solutions That Can Improve Evacuation Time

As a first proposed solution to reduce evacuation times in emergency situations, the construction of two new access roads to the educational institution are considered, which can be used only in case of catastrophic events, such as fires or earthquakes, and these new access roads will be, in fact, emergency exits. Next, a situation is presented in which two emergency exits will be added in the middle of the side halls; a strategic location, which will provide a balance between the distances between the school rooms and these exits. To observe the new access roads, which can only be used in crisis situations, we can see the Figure 16.



Figure 16. Emergency exits from the model (designed by the authors in NetLogo).

For the model with the two emergency exits considered, 400 simulations were performed again based on the same scenarios presented in the previous section, with the new results being highlighted in Table 4.

Scenario	Situation	Total Evacuation Time (seconds)	Total Number of Injured People	Impact on the Initial Type of Discharge
Baseline scenario	-	116.462	0.017	-24.06%
	А	111.990	0.020	-27.06%
Scenario I Variability of access paths	В	115.323	0.029	-26.9%
, , , , , , , , , , , , , , , , , , ,	С	120.985	0.037	-34.3%
Scenario II	А	113.865	0.042	-32.76%
The emergence of obstacles as a factor of uncertainty	В	136.193	0.079	-27.86%
Scenario III	А	101.713	0.015	-30.47%
Variation of children present	В	94.185	0.007	-31.25%

Table 4. Simulation results.

Analyzing Table 4, it is observed that the evacuation times decrease by at least 24% compared to the initial situation, in which there were no emergency exits built. Therefore, this solution is an extremely viable alternative for a faster and safer evacuation, and the costs for building exits and adjacent systems (signals, indicators) are RON 16,000 (EUR 3500), according to Prompt System [62]. Considering the possibility of serious injury to evacuees, the investment is very small considering that there is a risk that in the scenario of a disaster, the educational institution will pay compensation much higher than the cost of new emergency exits.

Another way to drastically reduce evacuation time is through school regulations to limit the capacity of a classroom to a maximum of 25 students, compared to 30 as before. Thus, this change will make it possible to reach an emergency evacuation time considerably shorter than that presented in the initial situation.

#### 8. Conclusions and Discussion

The problem of emergency evacuation has been frequently studied, with the aim of wanting to optimize it as much as possible. Unfortunately, due to the variability and dynamics of human nature, an optimal model which takes into account all the influences reflected in the total evacuation time is almost impossible to determine because there is not only a small range of cases that can be studied, but there is an infinity of possible situations. An example of this can be the placement of obstacles, with some areas having a lower density of obstacles than others, or the fact that some obstacles can even help to improve evacuation time, although they are perceived generally as a bad thing. It is also impossible for all individual features to be included in the models, as all people are different from each other by at least one characteristic, a difference that proves to be one of the main factors in making emergency evacuation models, along with the eventuality of the unforeseen.

The purpose of this study was the analysis of evacuation simulations in emergency situations of a school. The model that was used took into account the individual particularities of the people who took part in the evacuation, emphasizing the effects that form in a crowd when evacuating; effects such as the funnel effect, which is caused by the formation of bottlenecks around narrow areas. At the same time, the importance of an evacuation being as fast as possible, carried out in the safest way possible by those taking part in the simulation was also taken into account. Interpreting the results of the simulation on different scenarios, we highlighted some issues related to emergency evacuation. In fact, the advantages of this model also stand out, these being:

Manages to expose in a precise way the differences of the evacuation times depending on the opening of the access ways that are available during the simulations;

- Succeeds in taking into account the obstacles that appear during the simulations, outlining the differences that appear in the evacuation times;
- Agencies are categories of reality, they can be diversified in terms of individual characteristics (different speeds depending on the category, people with disabilities present).

It should be noted that, following the study carried out on the basis of the "Regina Maria" school, the shortest evacuation time in case of emergency, with a distribution of people according to real data, was present in the situation where only 65% of the places addressed to students were occupied. An evacuation time of about two and a quarter minutes is one that is below the optimal average of school evacuation times, but this time can be reduced by implementing measures such as the construction of at least two new access roads, which are to be used only in emergency situations.

To have a clearer view on the correlation between the factors that could impact the evacuation time and the actual duration of the evacuation, we decided to consider these factors one by one in our scenarios, as this would better explain their individual importance in the analysis, rather than having them all considered in other scenarios. In this paper, we took into account our thoughts about what may impact the evacuation time, and also after observing that there are not many studies conducted which included them, with these items being actually only some of the many elements behind one's behavior, as the "emotional" part of humans will never be deterministic with regards to the outer factors. As a justification for the aspects covered in this article, a number of factors were considered which could be estimated in order to make our analysis possible, but, at the current rate of the technology improvement, especially with regards to the AI area, guaranteed is the fact that, in the near future, estimable factors could be taken from our thoughts and what we think and these could represent the factors that we would want considered in our further research regarding this paper's topic.

The variety of elements that contribute to the modification of human behavior and their reactions implies the understanding and knowledge of complex adaptive systems. Between all these elements there is connectivity and interdependence, functioning far from equilibrium and emergence. These are just a few properties of complex adaptive systems, so understanding them can help influence people's decisions in evacuating themselves in the context of disaster scenarios. Agent-based modeling, as a working tool from a cybernetic perspective, contributes to understanding how agents (in our case, students) react in different contexts. In order to observe how the model proposed by us in NetLogo works, we made a video available at https://github.com/qfanee/Disaster-Evacuation-Process (accessed on 10 September 2021). Furthermore, the written programming code is at the same online address.

The model we propose and the continuous improvement of life safety systems, together with the environmental settings, can be considered sustainable practices for the future. We aim to capture in future research other aspects that can improve the safety of human life.

**Author Contributions:** Conceptualization, Ş.I. and I.N.; data curation, Ş.I., I.N. and N.C.; formal analysis: I.N. and Ş.I.; investigation: Ş.I., I.N. and N.C.; methodology: I.N., Ş.I. and N.C.; software, Ş.I. and I.N.; supervision, I.N. and N.C.; validation, Ş.I. and I.N.; visualization, Ş.I., I.N. and N.C.; writing—original draft, Ş.I.; writing—review and editing, I.N. and Ş.I. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

**Acknowledgments:** We thank the editor and anonymous referees who read the paper and provided helpful comments for improvements.

Conflicts of Interest: The authors declare no conflict of interest.

# References

- The World Bank. In Europe Disaster Risk Capital Romanians Turn to Technology. Available online: https://www.worldbank.org/ ro/news/feature/2019/11/25/in-europe-disaster-risk-capital-romanians-turn-to-technology (accessed on 10 September 2021).
   European Commission. EU Regional Policy and Urban Development. Available online: https://ec.europa.eu/regional\_policy/
- en/projects/Romania/vision-2020-romania-improves-its-disaster-management-capacity (accessed on 10 September 2021).
- The World Bank. Global Economic Prospects. Available online: https://www.worldbank.org/en/publication/global-economicprospects (accessed on 12 September 2021).
- 4. European Commission. EU Global Response to COVID-19. Available online: https://ec.europa.eu/international-partnerships/ topics/eu-global-response-covid-19\_en#header-2844 (accessed on 12 September 2021).
- 5. Chen, W.; Fang, Y.; Zhai, Q.; Wang, W.; Zhang, Y. Assessing Emergency Shelter Demand Using POI Data and Evacuation Simulation. *ISPRS Int. J. Geo-Inf.* 2020, *9*, 41. [CrossRef]
- 6. Dewan, A.M. Hazards, Risk, and Vulnerability in Floods in a Megacity; Springer: Dordrecht, The Netherlands, 2013. [CrossRef]
- Lee, Y.; Keum, H.; Han, K.; Hong, W.H. A hierarchical flood shelter location model for walking evacuation planning. *Environ. Hazards* 2021, 20, 432–455. [CrossRef]
- 8. Shendarkar, A.; Vasudevan, K.; Lee, S.; Son, Y. Crowd simulation for emergency response using BDI agents based on Immersive Virtual Reality. *Simul. Model. Pract. Theory* **2008**, *16*, 1415–1429. [CrossRef]
- 9. LEED Program. Available online: https://www.usgbc.org/leed/why-leed (accessed on 24 February 2021).
- 10. BREEAM Projects. Available online: https://tools.breeam.com/projects/ (accessed on 24 February 2021).
- 11. RFI Romania. Available online: https://www.rfi.ro/social-130047-incendii-produse-spitalele-din-romania (accessed on 24 February 2021).
- 12. Bloomberg. Available online: https://www.bloomberg.com/news/articles/2021-01-29/fire-at-romania-s-largest-covid-19 -hospital-kills-four-patients (accessed on 24 February 2021).
- 13. BBC News. Available online: https://www.bbc.com/news/world-europe-34684973 (accessed on 24 February 2021).
- 14. Merkle, D.; Middendorf, M. Modeling the Dynamics of Ant Colony Optimization. Evol. Comput. 2002, 10, 235–262. [CrossRef] [PubMed]
- 15. Nica, I.; Chiriță, N. Holistic Approach of Complex Adaptive Systems. Theory, Application and Case Studies; Lambert Academic Publishing: Chisinau, Republic of Moldova, 2021; pp. 12–24, ISBN 978-620-3-30709-2.
- 16. Mauboussin, M.; Sullivan, T. Embracing Complexity. Harvard Business Publishing. Available online: https://hbsp.harvard.edu/ product/R1109E-PDF-ENG (accessed on 5 September 2021).
- 17. Akopov, A.; Beklaryan, G. Modelling the dynamics of the "Smarter Region". In Proceedings of the 2014 IEEE Conference on Computational Intelligence for Financial Engineering & Economics (CIFEr), London, UK, 27–28 March 2014.
- 18. Joo, J.; Kim, N.; Richard, W.; Ling, R.; Young-Jun, S.; Yeong-Gwang, O.; Lee, S. Agent-based simulation of affordance-based human behaviors in emergency evacuation. *Simul. Model. Pract. Theory* **2013**, *32*, 99–115. [CrossRef]
- 19. Lu, T.; Mingyuan, H.; Hui, L. Agent-based simulation of building evacuation: Combining human behavior with predictable spatial accessibility in a fire emergency. *Inf. Sci.* 2015, 295, 53–66.
- 20. Wang, Z.; Chen, T. Pedestrian Evacuation Modelling with Dynamics Congestion Avoidance. In Proceedings of the 9th International Conference on Pedestrian and Evacuation Dynamics (PED2018), Lund, Sweden, 21–24 August 2018.
- 21. Wang, J.; Zhang, L.; Shi, Q.; Yang, P.; Hu, X. Modeling and simulating for congestion pedestrian evacuation with panic. *Phys. A Stat. Mech. Its Appl.* **2015**, 428, 396–409. [CrossRef]
- 22. Sime, J. Crowd psychology and engineering. Saf. Sci. 1995, 21, 1–14. [CrossRef]
- 23. Guan, J.; Wang, K. Cooperative evolution in pedestrian room evacuation considering different individual behaviors. *Appl. Math. Comput.* **2019**, *369*, 124865. [CrossRef]
- 24. Wu, F.; Wang, G.; Si, Y.; Lin, P. An Experimental Study on the Exit Location on the Evacuation Efficiency under High Competition Condition. *Procedia Eng.* 2018, 211, 801–809. [CrossRef]
- 25. Wolshon, B.; Urbina, E.; Wilmot, C.; Levitan, M. Review of policies and practices for hurricane evacuation. I: Transportation planning, preparedness, and response. *Nat. Hazards Rev.* 2005, *6*, 129–142. [CrossRef]
- 26. Li, Z. An emergency exits choice preference model based on characteristics of individual diversity. *Adv. Mech. Eng.* 2017, *9*, 168781401769354. [CrossRef]
- 27. Faroqi, H.; Mesgari, M.-S. Agent-Based Crowd Simulation Considering Emotion Contagion for Emergency Evacuation Problem. ISPRS-Int. Arch. Photogramm. Remote Sens. Spat. Inf. Sci. 2015, 40, 193–196. [CrossRef]
- 28. Zhang, G.; Lu, D.; Lv, L.; Yu, H.; Liu, H. Knowledge-Based Crowd Motion for the Unfamiliar Environment. *IEEE Access* 2018, 6, 72581–72593. [CrossRef]
- 29. Song, X.; Zhang, Z.; Peng, G.; Shi, G. Effect of authority figures for pedestrian evacuation at metro stations. *Phys. A Stat. Mech. Its Appl.* **2017**, *465*, 599–612. [CrossRef]

- 30. Helbing, D.; Johansson, A.; Al-Abideen, H.Z. Dynamics of crowd disasters: An empirical study. *Phys. Rev. E* 2007, 75, 046109. [CrossRef] [PubMed]
- 31. Jiang, G.; Ma, F.; Shang, J.; Chau, P.Y.K. Evolution of knowledge sharing behavior in social commerce: An agent-based computational approach. *Inf. Sci.* 2014, 278, 250–266. [CrossRef]
- 32. Vermuyten, H.; Belien, J.; De Boeck, L.; Reniers, G. A review of optimisation models for pedestrian evacuation and design. *Saf. Sci.* **2016**, *87*, 167–175. [CrossRef]
- Duives, D.; Daamen, W.; Hoogendoorn, S. State-of-the-art crowd motion simulation models. *Transp. Res. Part C Emerg. Technol.* 2013, 37, 193–209. [CrossRef]
- 34. Wood, G. A Survey of Behavior in Fires. In *Fires and Human Behaviour;* Canter, D., Ed.; John Wiley & Sons: New York, NY, USA, 1980; pp. 83–95.
- 35. Sime, J. *Report No. 45 Human Behaviour in Fires Summary Report;* Building Use & Safety Research Unit, School of Architecture, Portsmonth Polytechnic: London, UK, 1992; 385p.
- 36. Proulx, G. Occupant Behaviour and Evacuation. In Proceedings of the 9th International Fire Protection Symposium, Munich, Germany, 25–26 May 2001; pp. 219–232.
- 37. Heba, K.; Al-Megren, S.; Althunyan, R.; Almulifi, A. Effect of exit placement on evacuation plans. *Eur. J. Oper. Res.* 2018, 269, 749–759. [CrossRef]
- Industry Safe. Understanding the Importance of Emergency Exit Routes. Available online: https://www.industrysafe.com/blog/ safety-management/understanding-the-importance-of-emergency-exit-routes (accessed on 1 March 2021).
- 39. Kaneberg, E. Emergency Preparedness Management and Civil Defence in Sweden: An All-Hazards Approach for Developed Countries' Supply Chains. Ph.D. Thesis, Jönköping International Business School, Jönköping, Sweden, 2018. Available online: http://www.diva-portal.org/smash/get/diva2:1206379/FULLTEXT01.pdf (accessed on 1 March 2021).
- 40. Heath, M.; Ryan, K.; Dean, B.; Bingham, R. History of School Safety and Psychological First Aid for Children. *Brief Treat. Crisis Interv.* 2007, 7, 206–223. [CrossRef]
- 41. Ronan, K.; Alisic, E.; Towers, B.; Johnson, V.; Johnston, D. Disaster Preparedness for Children and Families: A Critical Review. *Curr. Psychiatry Rep.* **2015**, *17*, 589. [CrossRef]
- Hayden, J.; Petal, M. Child-Centred Risk Reduction Research-into-Action Brief: Early Childhood and Disaster Risk Reduction, School Emergency Drills. 2018. Available online: <a href="https://reliefweb.int/sites/reliefweb.int/files/resources/61523\_earlychildhoodanddrrr2abriefeng2018.pdf">https://reliefweb.int/sites/reliefweb.int/sites/reliefweb.int/files/resources/61523\_ earlychildhoodanddrrr2abriefeng2018.pdf</a> (accessed on 1 March 2021).
- 43. NaviGate Prepared. Improving School Safety with Enhanced Drills. 2017. Available online: https://cdn2.hubspot.net/hubfs/58 43129/Safety%20Drills/SafetyDrills\_WhitePaper.pdf (accessed on 26 February 2021).
- 44. Wilensky, U.; Rand, W. An Introduction to Agent-Based Modeling; The MIT Press: London, England, 2016.
- 45. Railsback, S.; Grimm, V. Agent-Based and Individual-Based Modeling: A Practical Introduction, Second edition; Princeton University Press: Princeton, NJ, USA, 2019.
- Bradshaw, J.; Feltovich, P.; Johnson, M. Human-Agent Interaction. In *Handbook of Human-Machine Interaction*; Boy, G., Ed.; CRC Press: Boca Raton, FL, USA, 2011; pp. 293–302.
- 47. Xiaoshan, P.; Charles, H.; Kincho, H. A Multi-Agent based simulation framework for the study of human and social behavior in Egress analysis. *Comput. Civ. Eng.* **2005**. [CrossRef]
- 48. De Marchi, S.; Page, S. Agent-Based Models. Anu. Rev. Political Sci. 2014, 17, 1–20. [CrossRef]
- 49. Kirman, A. The economy as an interactive system. In *The Economy as an Evolving Complex System II.*; CRC Press: Boca Raton, FL, USA, 1997.
- 50. Vriend, N. An illustration of the essential difference between individual and social learning, and its social consequences for computational analyses. *J. Econ. Dyn. Control* 2000, 24, 1–19. [CrossRef]
- 51. LeBaron, B. Stochastic volatility as a simple generator of apparent financial power laws and long memory. *Quant. Financ.* **2001**, *1*. [CrossRef]
- 52. Miller, J.; Page, S. Complex Adaptive Systems: An Introduction to Computational Models of Social Life; Princeton University Press: Princeton, NJ, USA, 2007.
- 53. Bruch, E.; Atwell, J. Agent-Based Models in Empirical Social Research. Sociol. Methods Res. 2013, 44. [CrossRef]
- 54. National Institute of Statistics. Available online: http://statistici.insse.ro/ (accessed on 28 February 2020).
- 55. Thiele, J.; Kurth, W.; Grimm, V. Facilitating Parameter Estimation and Sensitivity Analysis of Agent-Based Models: A Cookbook Using NetLogo and R. *J. Artif. Soc. Soc. Simul.* **2014**, *17*. [CrossRef]
- Peacock, R.; Hoskins, B.; Kuligowski, E. Overall and Local Movement Speeds during Fire Drill Evacuations in Building up to 31 Stories. NIST Technical Note 1675, 2010. Available online: https://tsapps.nist.gov/publication/get\_pdf.cfm?pub\_id=906798 (accessed on 28 February 2021).
- 57. Peacock, R.; Kuligowski, E.; Averill, J. Pedestrian and Evacuation Dynamics. Springer: Gaithersburg, MD, USA, 2011.
- 58. Kawasaki, H.; Yamasaki, S.; Rahman, M.; Murata, Y.; Iwasa, M.; Teramoto, C. Teachers-parents cooperation in disaster preparation when schools become as evacuation centers. *Int. J. Disaster Risk Reduct.* **2020**, *44*. [CrossRef]
- 59. Chu, H.; Yu, J.; Wen, J.; Yi, M.; Chen, Y. Emergency evacuation simulation and management optimization in Urban Residential Communities. *Sustainability* **2019**, *11*, 795. [CrossRef]

- 60. Wei, Y.-F.; Shi, W.; Song, T. Approach to Effect of Obstacle on Pedestrian Evacuation with a Small-Grid Lattice Gas Model. *Int. Procedia Eng.* **2012**, *31*, 1077–1082. [CrossRef]
- 61. Safelenics. What Benchmark of Length for Evacuation Should Be Used? Available online: www.safelincs-forum.co.uk (accessed on 26 February 2021).
- 62. PromptSystem. Available online: https://www.promptsystem.ro/ (accessed on 1 March 2021).